Author's response to reviews

Title: Understanding older women's decision making and coping in the context of breast cancer treatment.

Authors:

Kate J Lifford (Liffordkj@cardiff.ac.uk)
Jana Witt (Jana.Witt@kcl.ac.uk)
Maria Burton (M.Burton@shu.ac.uk)
Karen Collins (K.Collins@shu.ac.uk)
Lisa Caldon (L.caldon@sheffield.ac.uk)
Adrian Edwards (EdwardsAG@cardiff.ac.uk)
Malcolm Reed (M.Reed@bsms.ac.uk)
Lynda Wyld (L.wyld@sheffield.ac.uk)
Kate Brain (BrainKE@cardiff.ac.uk)

Version: 2 Date: 13 January 2015

Author's response to reviews:

Line numbers have been added to the manuscript.